

# Novartis ILARIS® (canakinumab) receives FDA approval for new indication to treat Adult-Onset Still's Disease (AOSD)

Jun 16, 2020

- FDA granted an indication for active Still's disease including both systemic juvenile idiopathic arthritis (SJIA) and AOSD, a serious and rare inflammatory disorder with high unmet medical need<sup>1,2</sup>
- ILARIS (canakinumab) is the first FDA-approved treatment for AOSD
- AOSD has an estimated prevalence of less than 1 case per 100,000 people<sup>3</sup>

**East Hanover, June 16, 2020** - Novartis ILARIS (canakinumab) received US Food and Drug Administration (FDA) approval for a new indication to treat Adult-Onset Still's Disease (AOSD)<sup>1</sup>. The FDA granted an indication for active Still's disease including both systemic juvenile idiopathic arthritis (SJIA) and AOSD in patients aged 2 years and older<sup>1</sup>.

"At Novartis, we are committed to bringing medicines that address high unmet needs to patients, including to those who are living with rare diseases," said Victor Bultó, President, Novartis Pharmaceuticals Corporation. "We are very proud that ILARIS is the first treatment approved by the FDA for patients with AOSD, a serious and rare inflammatory disorder."

The efficacy of ILARIS in adults with AOSD is based on the pharmacokinetic exposure and extrapolation of the established efficacy of ILARIS in SJIA patients<sup>1</sup>. Efficacy of ILARIS was also assessed in a randomized, double-blind, placebo-controlled study that enrolled 36 patients (22 to 70 years old) diagnosed with AOSD<sup>4</sup>. The efficacy and safety data in AOSD were generally consistent with the results of a pooled analysis of SJIA patients<sup>1</sup>.

# INDICATION

ILARIS® (canakinumab) is indicated for the treatment of active Still's disease, including Adult-Onset Still's Disease (AOSD) and Systemic Juvenile Idiopathic Arthritis (SJIA) in patients aged 2 years and older.

# IMPORTANT SAFETY INFORMATION

# CONTRAINDICATION

ILARIS is contraindicated in patients with confirmed hypersensitivity to the active substance or to any of the excipients.

# WARNINGS AND PRECAUTIONS

#### **Serious Infections**

ILARIS has been associated with an increased risk of serious infections. Physicians should exercise caution when administering ILARIS to patients with infections, a history of recurring infections or underlying conditions, 1/5

which may predispose them to infections.

ILARIS should not be administered to patients during an active infection requiring medical intervention. Administration of ILARIS should be discontinued if a patient develops a serious infection.

Infections, predominantly of the upper respiratory tract, in some instances serious, have been reported with ILARIS. Generally, the observed infections responded to standard therapy. Isolated cases of unusual or opportunistic infections (eg, aspergillosis, atypical mycobacterial infections, cytomegalovirus, herpes zoster) were reported during ILARIS treatment. A causal relationship of ILARIS to these events cannot be excluded. In clinical trials, ILARIS has not been administered concomitantly with Tumor Necrosis Factor (TNF) inhibitors. An increased incidence of serious infections has been associated with administration of another interleukin-1 (IL-1) blocker in combination with TNF inhibitors. Coadministration of ILARIS with TNF inhibitors is not recommended because this may increase the risk of serious infections.

Drugs that affect the immune system by blocking TNF have been associated with an increased risk of new tuberculosis (TB) and reactivation of latent TB. It is possible that use of IL-1 inhibitors, such as ILARIS, increases the risk of reactivation of TB or of opportunistic infections.

Prior to initiating immunomodulatory therapies, including ILARIS, patients should be evaluated for active and latent TB infection. Appropriate screening tests should be performed in all patients. ILARIS has not been studied in patients with a positive TB screen, and the safety of ILARIS in individuals with latent TB infection is unknown. Patients testing positive in TB screening should be treated by standard medical practice prior to therapy with ILARIS. All patients should beinstructed to seek medical advice if signs, symptoms, or high risk exposure suggestive of TB (eg, persistent cough, weight loss, subfebrile temperature) appear during or after ILARIS therapy.

# **Immunosuppression**

The impact of treatment with anti-IL-1 therapy on the development of malignancies is not known. However, treatment with immunosuppressants, including ILARIS, may result in an increase in the risk of malignancies.

# **Hypersensitivity**

Hypersensitivity reactions have been reported with ILARIS therapy. During clinical trials, no anaphylactic reactions attributable to treatment with canakinumab have been reported. It should be recognized that symptoms of the underlying disease being treated may be similar to symptoms of hypersensitivity. If a severe hypersensitivity reaction occurs, administration of ILARIS should be discontinued and appropriate therapy initiated.

# **Immunizations**

Live vaccines should not be given concurrently with ILARIS. Prior to initiation of therapy with ILARIS, patients should receive all recommended vaccinations. In addition, because ILARIS may interfere with normal immune response to new antigens, vaccinations may not be effective in patients receiving ILARIS.

Canakinumab, like other monoclonal antibodies, is actively transported across the placenta mainly during the third trimester of pregnancy and may cause immunosuppression in the *in utero* exposed infant. The risks and benefits should be considered prior to administering live vaccines to infants who were exposed to ILARIS *in utero* for at least 4 to 12 months following the mother's last dose of ILARIS.

# **Macrophage Activation Syndrome**

Macrophage Activation Syndrome (MAS) is a known, life-threatening disorder that may develop in patients with rheumatic conditions, in particular Still's disease, and should be aggressively treated. Physicians should be attentive to symptoms of infection or worsening of Still's disease, as these are known triggers for MAS. Eleven cases of MAS were observed in 201 SJIA patients treated with canakinumab in clinical trials. Based on the clinical trial experience, ILARIS does not appear to increase the incidence of MAS in Still's disease patients, but no definitive conclusion can be made.

#### **ADVERSE REACTIONS**

The most common adverse drug reactions greater than 10% associated with ILARIS treatment in SJIA patients were infections (nasopharyngitis and upper respiratory tract infections), abdominal pain, and injection site reactions.

Please see full Prescribing Information, including Medication Guide, for ILARIS.

#### **About Novartis**

Located in East Hanover, NJ Novartis Pharmaceuticals Corporation – an affiliate of Novartis – is reimagining medicine to improve and extend people's lives. As a leading global medicines company, we use innovative science and digital technologies to create transformative treatments in areas of great medical need. In our quest to find new medicines, we consistently rank among the world's top companies investing in research and development. Novartis employs about 15,000 people in the United States. For more information, please visit <a href="http://www.novartis.us">http://www.novartis.us</a>.

Novartis and Novartis US is on Twitter. Sign up to follow @Novartis at <a href="http://twitter.com/novartisnews">http://twitter.com/novartisnews</a> and @NovartisUS at <a href="https://twitter.com/NovartisUS">https://twitter.com/NovartisUS</a>

For Novartis multimedia content, please visit www.novartis.com/news/media-library

For questions about the site or required registration, please contact media.relations@novartis.com

###

# **Novartis Media Relations**

E-mail: media.relations@novartis.com

#### **Eric Althof**

#### Jeannie Neufeld

Head, US Corp & Country External Comms Director, Communications +1 646 438 4335 +1 862 778 2104

<u>eric.althoff@novartis.com</u> <u>jeannie.neufeld@novartis.com</u>

# **Jamie Bennett**

Director, US Media Relations +1 862 217 3976 jamie.bennett@novartis.com

#### **Novartis Investor Relations**

E-mail: <u>investor.relations@novartis.com</u>

3/5

# **Sloan Simpson Cory Twining**

+1 862 778 5052 +1 862 778 3258

#### References

- 1. ILARIS [Prescribing Information]. East Hanover, NJ: Novartis Pharmaceuticals Corp; 2020.
- 2. National Organization for Rare Disorders (NORD). Adult onset Still's disease. Updated 2015. Accessed May 5, 2020. <a href="https://rarediseases.org/rare-diseases/adult-onset-stills-disease/">https://rarediseases.org/rare-diseases/adult-onset-stills-disease/</a>
- 3. Efthimiou P. Auto-Inflammatory Syndromes. Springer Nature Switzerland AG; 2019.
- 4. Kedor C, et al. Canakinumab for treatment of Adult-Onset Still's Disease to achieve reduction of arthritic manifestation (CONSIDER): phase II, randomised, double-blind, placebo-controlled, multicentre, investigator-initiated trial. *Ann Rheum Dis.* 2020;0:1-8.

# **Disclaimer**

This media update contains forward-looking statements within the meaning of the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements can generally be identified by words such as "potential," "can," "will," "plan," "may," "could," "would," "expect," "anticipate," "seek," "look forward," "believe," "committed," "investigational," "pipeline," "launch," or similar terms, or by express or implied discussions regarding potential marketing approvals, new indications or labeling for the investigational or approved products described in this media update, or regarding potential future revenues from such products. You should not place undue reliance on these statements. Such forward-looking statements are based on our current beliefs and expectations regarding future events, and are subject to significant known and unknown risks and uncertainties. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those set forth in the forward-looking statements. There can be no guarantee that the investigational or approved products described in this media update will be submitted or approved for sale or for any additional indications or labeling in any market, or at any particular time. Nor can there be any guarantee that such products will be commercially successful in the future. In particular, our expectations regarding such products could be affected by, among other things, the uncertainties inherent in research and development, including clinical trial results and additional analysis of existing clinical data; regulatory actions or delays or government regulation generally; global trends toward health care cost containment, including government, payor and general public pricing and reimbursement pressures and requirements for increased pricing transparency; our ability to obtain or maintain proprietary intellectual property protection; the particular prescribing preferences of physicians and patients; general political, economic and business conditions, including the effects of and efforts to mitigate pandemic diseases such as COVID-19; safety, quality, data integrity or manufacturing issues; potential or actual data security and data privacy breaches, or disruptions of our information technology systems, and other risks and factors referred to in Novartis AG's current Form 20-F on file with the US Securities and Exchange Commission. Novartis is providing the information in this media update as of this date and does not undertake any obligation to update any forward-looking statements contained in this media update as a result of new information, future events or otherwise.

**Source URL:** https://www.novartis.com/us-en/news/media-releases/novartis-ilaris-canakinumab-receives-fda-approval-new-indication-treat-adult-onset-stills-disease-aosd

# List of links present in page

- https://www.novartis.com/us-en/us-en/news/media-releases/novartis-ilaris-canakinumab-receives-fda-approval-new-indication-treat-adult-onset-stills-disease-aosd
- https://www.novartis.com/us-en/us-en/sites/novartis\_us/files/2022-03/ilaris\_0.pdf

- https://www.novartis.com/us-en/us-en/home
- https://twitter.com/novartisnews
- https://twitter.com/NovartisUS
- https://www.novartis.com/news/media-library
- mailto:media.relations@novartis.com
- mailto:eric.althoff@novartis.com
- mailto:jeannie.neufeld@novartis.com
- mailto:jamie.bennett@novartis.com
- mailto:investor.relations@novartis.com
- https://rarediseases.org/rare-diseases/adult-onset-stills-disease/